Author's response to reviews

Title: Use Of The Evidence Base In Substance Abuse Treatment Programs for American Indians and Alaska Natives: Pursuing Quality in the Crucible Of Practice And Policy

Authors:

Douglas K Novins (douglas.novins@ucdenver.edu)
Gregory A Aarons (gaarons@ucsd.edu)
Sarah G Conti (samiam@pagosa.net)
Dennis Dahlke (dahlke@frontier.net)
Raymond Daw (ray.daw@nndoh.org)
Alexandra Fickenscher (alexandra.fickenscher@ucdenver.edu)
Candace Fleming (candace.fleming@ucdenver.edu)
Craig Love (CraigLove@westat.com)
Kathleen Masis (epimasis@mtwytlc.com)
Paul Spicer (paul.spicer@ou.edu)

Version: 9 Date: 15 June 2011

Author's response to reviews:

The proofed manuscript looks great. My only recommendation is that there is a bold horizontal line in Table 2, just above "Sources of communication and influence. Uptake of innovation influenced by:". This should probably be a non-bold solid line. Thank you for your assistance. - Douglas Novins